Author's response to reviews

Title: Assessing effects of a media campaign on HIV/AIDS awareness and prevention in Nigeria: results from the VISION Project

Authors:

Joseph Keating (jkeating@tulane.edu)
Dominique Meekers (dmeekers@tulane.edu)
Alfred Adewuyi (alfredadewuyi@yahoo.com)

Version: 6 Date: 1 May 2006

Author's response to reviews:

May 01, 2006

The Editor
BMC Public Health

Re: Manuscript ID 1438657617884574 Proofing-stage

Dear Sir or Madam,

Attached is the final revision of the manuscript entitled 'Assessing effects of a media campaign on HIV/AIDS awareness and prevention in Nigeria: results from the VISION Project'. Small editorial corrections have been made to the manuscript. All authors have participated in the drafting, editing, and final revision of this manuscript. We have uploaded the final revised version. We thank you for all consideration given to this manuscript, and look forward to its publication.

Regards,

Joseph Keating
Research Assistant Professor
Tulane University School of Public Health and Tropical Medicine
1440 Canal Street, Suite 2200
New Orleans, LA, 70112